Author's response to reviews

Title: Which outcome expectancies are important in determining young adults' intentions to use condoms with casual sexual partners?: A cross-sectional study

Authors:

Katie V Newby (k.newby@coventry.ac.uk)
Katherine E Brown (k.brown@coventry.ac.uk)
David P French (D.French@coventry.ac.uk)
Louise M Wallace (LWallace@cad.coventry.ac.uk)

Version: 2 Date: 30 April 2012

Author's response to reviews: see over
30th April 2012

Natalie Pafitis
Executive editor
BMC Public Health

Mrs Katie Newby
ARC Health and Lifestyle Interventions
WF121, School of Health and Life Sciences
Coventry University
Priory Street, Coventry
CV1 5FB

Dear Ms Pafitis,

Re: Resubmission of manuscript to BMC Public Health

I wish to resubmit the paper ‘Which outcome expectancies are important in determining young adults’ intentions to use condoms with casual sexual partners?: A cross-sectional study’ for consideration of publication within BMC Public Health.

Following an initial assessment of our paper, I have made the following amendments:

- Clarified from whom written informed consent was obtained (p10)
- Included a title page at the front of the manuscript

I hope that you find the paper interesting and look forward to hearing from you in due course.

Yours sincerely

Katie Newby BSc (Hons), MSc
Research Fellow
Applied Research Centre in Health and Lifestyle Interventions (ARC-HLI)